Author’s response to reviews

Title: Low dose of flurbiprofen axetil decrease the rate of acute kidney injury after operation: a retrospective clinical data analysis of 9915 cases

Authors:

Dong Wang (905486084@qq.com)
Shi-Kun Yang (yangshikun@csu.edu.cn)
Meng-Xi Zhao (249619441@qq.com)
Yong-Zhong Tang (tangyongzong@csu.edu.cn)
Wen OU-Yang (ouyangwen133@vip.sina.com)
Hao Zhang (zhanghaoliaoqing@163.com)
Qin Liao (xy3yliaoqin@sina.com)

Version: 2 Date: 21 Jan 2020

Author’s response to reviews:

Dear Dr Charat Thongprayoon,

We respectfully send back our manuscript: Low dose of flurbiprofen axetil decrease the rate of acute kidney injury after operation: a retrospective clinical data analysis of 9915 cases (Manuscript-ID: BNEP-D-19-00829.R2) for the consideration of publication in the Journal of BMC Nephrology.

We greatly appreciate the efforts of the reviewers for the valuable and constructive comments on previous draft. Based on their comments and requests, we have carefully taken their comments into consideration and made extensive modification on the original manuscript. The following summarizes how we responded to reviewer comments. A revised manuscript with the correction sections marked was attached as the supplemental material and for the purpose of easy check.

Here is our response to their comments.

Revision — authors’ response
Reviewer #1
Comment-1:
1. All my concerns were addressed. I have no further reservations for the publication of this revised version.
Response: Thanks for your valuable comments.

Reviewer #2
Comment-1:
1. The title of the article is "Low dose of flurbiprofen axetil decrease the rate of acute kidney injury after operation: a retrospective clinical data analysis of 9915 cases". In the revised manuscript, the authors convincingly enough replied to my previous comments and properly improved the manuscript. The manuscript contains now all information.
Response: Thanks for your valuable comments.

Reviewer #3
The manuscript substantially improved after revision. I have a few minor comments.
Comment-1: The manuscript might be too long (4485 words), particularly in the section regarding comparison of OR by multivariate regression analysis and ordinal logistic regression analysis. Please revise the paragraphs.
Response: Thanks for your valuable comments, we have carefully checked the whole draft and revised some relevant content, the revised manuscript was about 3200 words, please check it, thanks a lot.

Comment-2: Background: line 27 .. "a 52-week" should be used instead of "a 52-weeks"
Response: We have amended the relevant content in the chapter of Background, please check it, thanks a lot.

Comment-3. Page 11 line 24 ... "Risk factors" should be used instead of "risk factor".
Response: This is a pertinent advice, we have modified “risk factor” to “risk factors”, please check it. In addition, we have carefully checked the whole draft, thanks a lot.

We will give our heartfelt thanks for the reviewers’ and the editors’ comments, and we believe that the amendments will make this manuscript more valuable. We hope the reviewers and the editors will be satisfied with the revisions for the original manuscript. If you have any question about this paper, please don’t hesitate to let us know.

Thanks and best regards!
Yours sincerely,

Dr. Qin Liao,
Department of Anesthesiology
The Third Xiangya Hospital, Central South University
No.138, Tongzipo Road, Changsha, Hunan Province, China
Jan 18, 2020